Author's response to reviews

Title: Favourable ten-year overall survival in a Caucasian population with high probability of hereditary breast cancer

Authors:

Laura Cortesi (hbc@unimo.it)
Cristina Masini (cristinamas@libero.it)
Claudia Cirilli (cirilli.claudia@policlinico.mo.it)
Veronica Medici (veronica.medici@gmail.com)
Isabella Marchi (robertoisa@alice.it)
Giovanna Cavazzini (giovanna.cavazzini@ospedalimantova.it)
Giuseppe Pasini (Giuseppe.Pasini@auslrn.net)
Daniela Turchetti (danielat@med.unibo.it)
Massimo Federico (mfederico@unimore.it)

Version: 8 Date: 8 March 2010

Author's response to reviews:

In the title caucasian has been changed in Caucasian.
Hematology has been changed in Haematology
Mantova has been changed in Mantua
In the abstract session (Background)BRCA was changed in BRCA1
Fig.1 has been modified